Author's response to reviews

Title: Validation study of a diagnostic aid to rule out pneumonia in adults with cough and fever

Authors:

Ulrike Held (ulrike.held@usz.ch)
Claudia Steurer-Stey (claudia.stey@usz.ch)
Felix Huber (felix.huber@medix.ch)
Sergio Dallafior (dallafior@pg-altstetten.ch)
Johann Steurer (johann.steurer@usz.ch)

Version: 3  Date: 19 August 2012

Author's response to reviews:

Dear Nathaniel A. Nazareno,

according to your recommendations we made changes in the text (result section). We have removed age and sex of the three patients who were prescribed antibiotics during the study period.

amy thanks for giving us the opportunity to make these changes

best wishes

Johann Steurer